Author's response to reviews

Title: Socioeconomic position and health status of people who live near busy roads: the Rome Longitudinal Study (RoLS)

Authors:

Giulia Cesaroni (cesaroni@asplazio.it)
Chiara Badaloni (badaloni@asplazio.it)
Valeria Romano (valeria.romano@comune.roma.it)
Eugenio Donato (eugenio.donato@comune.roma.it)
Carlo A Perucci (perucci@asplazio.it)
Francesco Forastiere (forastiere@asplazio.it)

Version: 5 Date: 12 July 2010

Author's response to reviews:

Dear Editorial Team,

I changed few things:

Title page: all e-mail addresses are written in the same way (they all are links to the e-mail).
Page 4: r.8 After "In some of the studies" I added "evaluating heath effects".
Page 7: r.14 I changed "52% of all road segments" in "26% of all roads".
Page 23: I changed in the title of Figure 2 (HTR) in (HTR, >=10,000 vehicles/day) I will upload the same figures in a jpg format

Best regards

Giulia Cesaroni